Author's response to reviews

Title:A National Cross-sectional Study for Poliovirus Seroprevalence in the Republic of Korea in 2012: Deficiency in Immunity to Polio among Middle-aged People

Authors:

HyeJin Kim (limetree81@naver.com)
Seoyeon Hwang (0117772624@hanmail.net)
So Min Lee (somin6510@korea.kr)
Yunhyung Kwon (yhhodori@gmail.com)
Gwangsook Park (gspark@korea.kr)
Yong Joon Park (pahmun@hanmail.net)
Geun-Ryang Bae (bgr824@naver.com)
Yong-Seok Jeong (ysjeong@khu.ac.kr)
Sang Won Lee (epilsw@korea.kr)
Ji-Yeon Hyeon (jyhyeon1205@gmail.com)

Version:2 Date:9 December 2014

Author's response to reviews: see over
December 9, 2014

Jodie McVernon
Editor of Viral disease section
BMC Infectious Disease

Dear Dr. Jodie McVernon

I, along with my coauthors, would like to ask you to consider the attached manuscript entitled “A National Cross-sectional Study for Poliovirus Seroprevalence in the Republic of Korea in 2012: Deficiency in Immunity to Polio among Middle-aged People” for publication in the BMC Infectious Disease as a research article.

This is the first large-scale study to investigate the polio seroprevalence since 2004, when an inactivated polio vaccine was introduced in the Republic of Korea by a nationwide cross-sectional study. We believe that the findings of this study are relevant to the scope of your journal and will be of interest to its readership.

Thank you for your consideration. I look forward to hearing from you.

Sincerely,
Ji-Yeon Hyeon

Division of Vaccine Research
Korea National Institute of Health, Korea Centers for Disease Control and Prevention
187 OsongSangmyeong2-ro, Osong-eup
CheongJu, Chungbuk 363-951,
Republic of Korea
Phone: 82-43-719-8152
Fax: 82-43-719-8189
Email: jyhyeon1205@gmail.com